Author's response to reviews

Title: Pantoea spp. causing early onset neonatal sepsis (EONS): an unusual case report

Authors:

Shreekant Tiwari (drshreekant@rediffmail.com)
Sibashanker Beriha (drsiba@rediffmail.com)

Version: 3
Date: 24 December 2014

Author's response to reviews: see over
Respected sir

Herewith we are re-submitting the manuscript as per your advice for your kind consideration for peer review. Pantoea spp. causing neonatal sepsis is the first documented case report from India. Definitely its identification and antibiotic susceptibility pattern will help to the clinicians for the better management of neonatal sepsis. This is an original case report of this kind and will help to the physicians to understand its pathogenesis and management. Till now only 19 cases of Pantoea spp. causing neonatal sepsis had been reported in world literature and out of which only 4 cases from full term baby. This is very rare case report and first documented case from India.